Author's response to reviews

Title: Time trends in single versus concomitant neck and back pain in Finnish adolescents: results from national cross-sectional surveys from 1991 to 2011

Authors:

Minna Ståhl (minna.stahl@tampere.fi)
Ashraf El-Metwally (elmetwallyas@ngha.med.sa)
Arja Rimpelä (arja.rimpela@uta.fi)

Version: 7 Date: 27 August 2014

Author's response to reviews:

Dear Editor,

We are very sorry that the STROBE checklist was not completed in the last revision. This was because we didn't understand what did you mean by completing the checklist. Now it has been completed with ticking the relevant statements and typing N/A next to those that were not applicable.

As requested, we have also removed the tables as additional files and incorporated them into the manuscript, after the references.

On behalf of all authors,
Minna Ståhl